\chapter{intro}
\section{einleitung halt}

\chapter{problemstellung}
\section{Auswertung ortsbezogener Spiele}

\chapter{forschungsstand}
\section{Metriken sammeln}
locomotion-action %TODO citation Reducing Location Overhead in Educational Geogames

Geogames: Designing Location-Based Games from Classic Board Games\cite{1705427}
\cite{heinz2015agent}
\cite{schlieder2005geogames}
\section{umgang mit trajektorien}
\cite{zheng2011computing,Zheng_2015,tang_motif,Chen_2012,pattern_recog}
\section{statistische verfahren}
\section{log processing}
\cite{steinegger2016analyse,andreassen2015monitoring,yang2016aggregated,sanjappa2017analysis,matotek2017performance,wilkinson2017practical}
\section{...}
\section{toDOI...}
\begin{verbatim}
10.1175/BAMS-D-14-00234.1
10.1016/j.tourman.2017.02.021
10.1080/13683500.2016.1194813
10.1136/bjsports-2016-096103
10.3390/s16111962
10.1007/s40534-015-0079-x
10.1016/j.sbspro.2014.07.239
10.1016/j.sbspro.2013.08.038
10.1007/978-3-319-11469-9_8
10.3390/s140712990
10.1080/13658816.2015.1086924
10.1080/13658816.2012.692791
10.3390/s90403033
10.1016/j.landurbplan.2009.07.008
\end{verbatim}
\section{todo...}
\begin{verbatim}
@inproceedings{pafka2013places,
  title={Places as intersecting flows: Mapping urban morphologies, functional constellations and pedestrian rhythms},
  author={Pafka, Elek},
  booktitle={4th Global Conference: Space and Place. Mansfield College, Oxford},
  year={2013}
}
@inproceedings{stopher2002gps,
  title={GPS, GIS and personal travel surveys: an exercise in visualisation},
  author={Stopher, Peter and Bullock, Philip and Jiang, Qingjian},
  booktitle={25th Australasian Transport Research Forum Incorporating the BTRE Transport Policy Colloquium},
  year={2002}
}
@inproceedings{koshak2008analyzing,
  title={Analyzing pedestrian movement in mataf using gps and gis to support space redesign},
  author={Koshak, Nabeel and Fouda, Abdullah},
  year = {2008}
}
\end{verbatim}



\chapter{lösungsansatz}
\section{modulare auswahl von metriken}
\section{spiel unabhängig}
\section{log processing taugt nich}

\chapter{umsetzung}
\section{klassenstruktur}
\section{konfigurierbarkeit (log import)}
\section{...}

\chapter{evaluierung}
\section{wie viele metriken umsetzbar?}
\section{erweiterbarkeit}

\chapter{diskussion}

\section{dummies}
\cite{zheng2011computing}
\nomenclature[s]{$E=mc^2$}{Energie}
\begin{longtable}[H]{|p{0.2\textwidth}|p{0.2\textwidth}|p{0.2\textwidth}|}
\hline
A&B&C\\
\hline
\caption{Tabelle 1}
\label{tab:tab1}
\end{longtable}
\image{3cm}{logo.png}{Uni-Logo}{img:uni}